Author's response to reviews

Title: Limb girdle muscular dystrophy type 2G with myopathic-neurogenic motor unit potentials and a novel muscle image pattern

Authors:

Ana Cotta (cotta@sarah.br)
Julia F Paim (julia@sarah.br)
Antonio L Cunha-Junior (400564@sarah.br)
Rafael X Neto (9679@sarah.br)
Simone V Nunes (7713@sarah.br)
Monica M Navarro (400039@sarah.br)
Jaquelin Valicek (400585@sarah.br)
Elmano Carvalho (9230@sarah.br)
Lydia U Yamamoto (lydiauy@usp.br)
Camila F Almeida (camila.almeida@usp.br)
Shelida V Braz (12758@sarah.br)
Reinaldo I Takata (202911@sarah.br)
Mariz Vainzof (mvainzof@usp.br)

Version: 5
Date: 18 September 2014

Author's response to reviews:

Point-by-point response

Dear Dr. Magdalena Morawska
Executive Editor BMC Clinical Pathology

Thank you very much for the acceptance of our manuscript for publication in BMC Clinical Pathology.

Please find bellow all minor corrections asked by the reviewer

1-
Line 49: "early tibialis anterior sparing"
Line 55: "tibialis anterior sparing"
Lines 128: "tibialis anterior muscle was spared"
Line 349 - Figure 1 legend: "tibialis anterior muscle sparing"

2) The "Authors contributions" section was corrected: the adequate initials are RXN and RIT (lines 249 and 251).

3) The expression "facial muscle weakness" replaced "facial muscle commitment" (line 53).

4) The small table (Table 1) was added at the end of the text (line 366).
5) The expressions "pseudoneurogenic" and "pseudohypertrophy" were excluded from the text (lines 49, 80, 174, 349).

6) Minor spelling errors were corrected, such as: "corresponding" (line 23), "31 years old" (line 50), "29 years old" (line 86), "clinical examination included muscle strength testing" (line 93), "telethoninopathy" (line 143), "20 years old" (line 159), "muscle weakness was slight" (line 163), "predominant" (line 170), "neither nerve degeneration nor groups of necrotic fibers" (lines 183-184), "telethoninopathy" (line 222), "FKRP gene - fukutin related protein gene" (line 237), "advanced biceps femoris" (line 342), "muscle tissue shows autophagic vacuoles (arrows) with membrane debris and myelinoid figures" (line 358), "atrophic muscle fiber with myofibrillar disorganization and few residual sarcomeres" (line 360).

Yours faithfully,
Mariz Vainzof